Author's response to reviews

Title: Rationale and design: telephone-delivered behavioral skills interventions for blacks with type 2 diabetes

Authors:

Leonard E Egede (egedel@musc.edu)
Joni L Strom (stromjl@musc.edu)
Valerie L Durkalski (durkalsv@musc.edu)
Patrick D Mauldin (mauldinp@musc.edu)
William P Moran (moranw@musc.edu)

Version: 6 Date: 24 March 2010

Author's response to reviews:

Minor editorial changes were made. We also updated enrollment information and updated the link for one of the references.